Author’s response to reviews

Title: Clinical Experience with Tigecycline in the Treatment of Hospital-Acquired Pneumonia Caused by Multidrug Resistant Acinetobacter baumannii

Authors:

Yangang Zhou (zhouyangang@csu.edu.cn)
Xumin Chen (1120340591@qq.com)
Ping Xu (xuping1109@csu.edu.cn)
Yan Zhu (zhuyan0918@csu.edu.cn)
Kuangguo Wang (aledith@126.com)
Daxiong Xiang (xiangdaxiong@163.com)
Feng Wang (roitech@126.com)
Hoan Banh (Hoan@ualberta.ca)

Version: 4 Date: 01 Mar 2019

Author’s response to reviews:

Dear editors,

Thank you very much for your feedback on “Clinical Experience with Tigecycline in the Treatment of Hospital-Acquired Pneumonia Caused by Multidrug Resistant Acinetobacter baumannii”, I am very appreciated about the suggestion you made. The manuscript was revised in accordance with BioMed Central editorial policies and formatting guidelines. Here are the responses to the reviewers.

1. We note that the current submission contains some textual overlap with other previously published works, in particular:


This overlap mainly exists in the Methods section. While we understand that you may wish to express some of the same ideas contained in these publications, please be aware that we cannot condone the use of text from previously published work.

Please re-phrase this section to minimize overlap.
Response: We have revised the manuscript as you suggested minimizing the overlap.

2. In the Funding section, please also describe the role of the funding body in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.

Response: The funding bodies didn’t involve in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript. We described this in “declaration” part of the manuscript as you suggested.

3. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Response: Done.